Author’s response to reviews

Title: Oxygen Therapy in Patients with ST Elevation Myocardial Infarction based on the culprit vessel: Results from the Randomized Controlled SOCCER Trial

Authors:

Arash Mokhtari (arash.mokhtari.0561@med.lu.se)
Mahin Akbarzadeh (mahin.a.akbarzadeh@skane.se)
David Sparv (david.sparv@med.lu.se)
Pallonji Bhiladvala (pallonji.bhiladvala@med.lu.se)
Håkan Arheden (hakan.arheden@med.lu.se)
David Erlinge (david.erlinge@med.lu.se)
Ardavan Khoshnood (ardavan.khoshnood@med.lu.se)

Version: 3 Date: 05 Feb 2020

Author’s response to reviews:

Dear Editor,

We would hereby like to submit a revised version of our manuscript “Oxygen Therapy in Patients with ST Elevation Myocardial Infarction based on the culprit vessel: Results from the Randomized Controlled SOCCER Trial”.

We have made all the editorial changes suggested. Our replies to the comments are in Times New Roman italics and red font in the following pages of this letter.

Figure 3 is made by different images. All of the images are downloaded from Wikimedia Commons.

It is our hope that the manuscript can now be accepted for publication in BMC Emergency Medicine.

Yours sincerely,

Ardavan Khoshnood

Corresponding author
Division of Emergency Medicine
Skåne University Hospital at Lund
SE-221 85 Lund, Sweden
Telephone: +46 46 222 17 10 00
Fax: +46 46 046 211 09 08
Email: ardavan.khoshnood@med.lu.se

-- Please state in the cover letter whether the images depicted in figure 3 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:
http://commons.wikimedia.org/wiki/Main_Page

ANSWER: This have been added to the cover letter. As stated, all images in the figure are downloaded from Wikimedia commons.

-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. (fig 3)

ANSWER: Figure 3 has been added. It was added to the Discussion section.

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

ANSWER: We have added the following sentence: The funding bodies had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

--Please include a Keywords section below your abstract listing three to ten keywords representing the main content of the article.

ANSWER: Ten keywords have been added.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
ANSWER: Respected.